Author's response to reviews

Title: Exclusive breastfeeding of Swedish children and possible influence on development of obesity: a prospective cohort study.

Authors:

Karina Huus (huca@hhj.hj.se)
Jonas F Ludvigsson (jonas.ludvigsson@yahoo.com)
Karin Enskar (Karin.Enskar@hji.hj.se)
Johnny Ludvigsson (Johnny.Ludvigsson@lio.se)

Version: 2 Date: 9 May 2008

Author's response to reviews: see over
Dear editor, M Norton

Thank you for the review of our manuscript entitled "Exclusive breastfeeding of Swedish children and possible influence on development of obesity: a prospective cohort study". We appreciate the constructive comments from the reviewer’s and are pleased to re-submit the manuscript with revisions as suggested.

The tables are now integrated in the manuscript as the reviewers wish but if that is a problem I can send them separate.

We look forward to hearing from you again.

Sincerely,

Karina Huus, RN, PhD student
Jonas F. Ludvigsson, MD, PhD

Karin. Enskär,RN, PhD
Johnny Ludvigsson, MD, PhD

Corresponding address:
Karina Huus
Department of Nursing Science
School of Health Science, Jönköping University
P.O.Box 1026, S-551 11 Jönköping
Sweden
Tel: +46 36 10 12 34 Fax:+46 36 10 12 50
E-mail: karina.huus@hhj.hj.se